Author's response to reviews

Title: Ventriculitis due to Staphylococcus lugdunensis: a case series

Authors:

Teresa Spanu (tspanu@rm.unicatt.it)
Donato Rigante (drigante@gmail.com)
Gianpietro Tamburrini (tamburrini@libero.it)
Barbara Fiori (fiori@rm.unicatt.it)
Tiziana D'Inzeo (dinzeo@rm.unicatt.it)
Brunella Posteraro (posteraro@libero.it)
Domenico Policicchio (policicchio@libero.it)
Maurizio Sanguinetti (sanguinetti@yahoo.it)
Giovanni Fadda (fadda@yahoo.it)

Version: 4 Date: 20 November 2007

Author's response to reviews:

Dear Editor,

enclosed you will find a manuscript entitled "Ventriculitis caused by Staphylococcus lugdunensis: a case series" which we would like to submit for the possible publication in Journal of Medical Case Reports.

We have followed all the suggestions received to prepare this revised form of the manuscript before starting the review-process.

This manuscript is original and has never been published before: all the authors have read and approved the content of the paper and each one contributed significantly to the work.

We look forward to receiving the comments of your editorial staff.

Thank you very much in advance for your attention.

Sincerely yours,

Teresa Spanu